\documentclass[12pt]{article}
\pagestyle{empty}
\setlength{\topmargin}{-1in}
\setlength{\oddsidemargin}{-.3in}
\setlength{\textheight}{9in}
\setlength{\textwidth}{7.2in}

\begin{document}

\begin{center}
{\Large Math 111 - 2,4

Summer 1997}
\end{center}

\vspace{.2in}

\begin{tabular}{lll}
INSTRUCTOR: & Blake Thornton \\
& Office:  Building 122, \#102 \\
& Office Phone:  581-4027 \\
& Office hours: By appointment \\
& \hspace{.3in} (good times are before 10 and after 12) \\
& E.mail:  {\tt thornton@math.utah.edu} \\
\end{tabular}

\vspace{.3in}

HOMEWORK:  \parbox[t]{5in}
{Homework is assigned but not collected.
If it is not done, you will probably not know the material well enough to 
do well on the tests.
And, the homework assigned is only the minimum you need to do.
To learn all the material it may be necessary for you to do more.
}

\vspace{.2in}

TESTS and GRADING:  \parbox[t]{5in}
{We will have 3 midterm tests and a final.
The 3 midterms will be worth 100 points each.
The final will be worth 200 points.
This gives a total of 500 points.
I will be no stricter then 10 \% points per letter grade (90-100:A, 80-90:B, etc.)}

\vspace{.2in}

\begin{tabular}{llllll}
TEST DATES:
& & 1 & Friday & July 11 \\
& & 2 & Wednesday & July 23\\
& & 3 & Friday & August 8  \\
& sec 2 & Final & Friday & August 22 & 10-12 \\
& sec 4 & Final & Thursday & August 21 & 10-12 \\
\end{tabular}


\end{document}





